

## ANSWERING REVIEWERS

Nov 10, 2015

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 19712-revised manuscript).

Title: Prophylactic managements of hepatitis B viral infection in liver transplantation

Authors: Takashi Onoe, Hiroyuki Tahara, Yuka Tanaka, Hideki Ohdan

Name of Journal: World Journal of Gastroenterology

ESPS Manuscript NO: 19712

Manuscript type: TOPIC HIGHLIGHT

Reviewer's comment:

In different parts of the manuscript there are sentences which deserve a proper citation, but this has been missing. So authors are requested to cite the related references for those sentences which need proper citation. Concluding remarks and also perspective for future woks should be added to the end of the manuscript.

Response to Reviewer

We would like to thank the reviewer for accepting our review.

We added concluding remarks and perspective for future woks parts. Further, we added proper citations to some sentences and changed bibliography according to reviewer's suggestion.

The manuscript has been improved according to the suggestions of the reviewer.

Thank you again for reviewer's constructive comments.

Sincerely yours,

Takashi Onoe, M.D., Ph.D.

Head, Division of Applied Immunobiology, Institution for Clinical Research,  
National Hospital Organization Kure Medical Center/Chugoku Cancer Center,

Clinical Professor, Gastroenterological and Transplant Surgery, Applied Life  
Sciences, Institute of Biomedical & Health Sciences, Hiroshima University,